Author’s response to reviews

Title: Association between glycosylated hemoglobin A1c and bone biochemical markers in type 2 diabetic postmenopausal women: a cross-sectional study

Authors:
lianzi Wang (2457767318@qq.com)
Tao Li (limedical1974@126.com;2457767318@qq.com)
Jiaqing Liu (292416530@qq.com)
Xian Wu (1929839872@qq.com)
Huihui Wang (1791902835@qq.com)
Xuemei Li (1476631550@qq.com)
Enjun Xu (649385314@qq.com)
Qiuli Chen (464968218@qq.com)
Chuan Yan (2795250845@qq.com)
Huimin Li (1170525680@qq.com)
Yuanhong Xu (872785560@qq.com)
Wei Wei (wwei@ahmu.edu.cn)

Version: 3 Date: 03 Mar 2019

Author’s response to reviews:

Thank you so much for giving us the chance to revise our manuscript entitled “Association between glycosylated hemoglobin A1c and bone biochemical markers in type 2 diabetic of postmenopausal women: a cross-sectional study”. we have study the advice and revised according your request, a single, final, clean version of our manuscript was upload, the figure was also uploaded as separate files. we hope the revised manuscript could meet with approval.
Thank you very much!

Sincerely yours,

Lianzi Wang, Tao Li, Jiaqing Liu, Xian Wu, Huihui Wang, Xuemei Li, Enjun Xu, Qiuli Chen, Chuan Yan, Huimin Li, Yuanhong Xu and Wei Wei.